Author’s response to reviews

Title: Using reflexivity to enhance in-depth interviewing skills for the clinician researcher.

Authors:

Ruth P McNair (r.mcnair@unimelb.edu.au)
Angela Taft (a.taft@latrobe.edu.au)
Kelsey Hegarty (k.hegarty@unimelb.edu.au)

Version: 5 Date: 29 October 2008

Author’s response to reviews:

Dear Editor,
I have made further changes as follows:
1. I have removed the header on all pages and title on page one
2. I have made the author list consistent
3. I have changed the format of presenting email addresses
4. removed keywords
5. ensured that the section names and order are correct (removed the methods heading)
6. Moved the table to the end
7. altered the format of the table as suggested